Author's response to reviews

Title: Papillary thyroid carcinoma with massive metastasis in the uterine corpus: a case report

Authors:

Jian-hong Wang Dr. (2921473917@qq.com)
Jing Yu Dr. (309473481@qq.com)
Chun-ping Ning Dr. (ningchunping1222@163.com)
Yong-mei Sun Dr. (549019112@qq.com)
Shi-bao Fang Dr. (152081340@qq.com)

Version: 3 Date: 24 October 2013

Author's response to reviews: see over
Dear Editors and Reviewers,

First of all, we thank both the reviewers and the editors for their positive and constructive comments and suggestions. We have already revised the manuscript following the comments of the reviewers and editors. We also accepted the professional editing service from Edanz as the suggestion of the editors. All the revisions were highlighted in red.

We hope the revised version of the manuscript is acceptable for publication in your journal. Looking forward to hearing from you.

Shi-bao Fang

**Replies to the reviewers:**

1. **Reviewer 1: Reviewer's report: no comments; Level of interest: An article of importance in its field; Quality of written English: Acceptable**

   **Answer:** Thank you very much for your positive comments. We appreciate it very much.

2. **Reviewer's report: No revisions necessary; Level of interest: An article of importance in its field; Quality of written English: Acceptable**

   **Answer:** Thank you very much! We appreciate it very much really.

Replies to the editors:

1. **After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further. We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.**
Answer: Thank you for your kind advice. We have already revised this manuscript with the help from Edanz. All the revision were highlighted in red.

2. Please Authors' contribution: Please place the Authors' Contributions section after Competing interests.
Answer: Thank you very much. The Authors’ contribution section was already placed after the competing interest.